Author’s response to reviews

Title: The Feasibility and Acceptability of a Primary School-Based Programme Targeting Diet and Physical Activity: The PhunkyFoods Programme

Authors:

Pinki Sahota (p.sahota@leedsbeckett.ac.uk)

Meaghan Christian (m.s.christian@leedsbeckett.ac.uk)

Rhiannon Day (r.day@leedsbeckett.ac.uk)

Kim Cocks (kcstats@aol.com)

Version: 2 Date: 13 Aug 2019

Author’s response to reviews:

The responses to the reviewers comments are provided in a document attached.